## Vietnam

-----

* Economic restructuring and trade liberalization further drives economy towards wage work, service work, the manufacturing sectors.
* Structural changes drove poverty down in absolute terms, but leave those in vulnerable positions consistently at-risk of slipping into or worsening existing poverty.
* Economic inequality in Vietnam intersectional between ethnic minorities, rural populations, regional and gendered dimensions.
* Ethnic minorities increase in economic inequality, driven by worse returns on assets (human capital and land) and worse access to endowments (land and educational infrastructure).
* Environmental degradation and environmental shocks consistently worsen within-sector inequalities for ethnic minority and female population.
* Vietnam in vulnerable position to increasing exogenous shocks due to climate change, building capacity against which may require focus shift on risk management and preventative measures

-----

Vietnam's economy is now firmly in the third decade of ongoing economic reform (*Doi Moi*) as a market-based economy,
which lead to remarkable growth phases through opening the economy to international trade while,
seen over the bulk of its population, attempting to keep inequality rates managed through policies of controlling credit and reducing subsidies to state-owned enterprises [@Bui2019].

<!-- poor/poverty <40%; mention low social mobility: different social insurances [@Bui2019] -->
Poverty in Vietnam is marked by a drastic reduction in absolute terms over this time with some of the decline directly attributable to the liberalization of markets over the country's growth more generally [@WorldBank2012; @McCaig2011; @Le2022].
While the rate of decline slowed since the mid-2000s [@VASS2006; @VASS2011],
it continued declining in tandem with small income inequality decreases.
The overall income inequality decrease that Vietnam experienced from the early 2000s suggests that economic growth has been accompanied by equity extending beyond poverty reduction [@Benjamin2017].
On the other hand, Le et al. [-@Le2021] suggest a slight increase in overall income distribution from 2010-2018.
At the same time, the population groups most affected by poverty through welfare inequalities stay unaltered, as do largely the primary factors accompanying it:
There is severe poverty persistence among ethnic minorities in Vietnam [@Baulch2012],
concomitant with low education and skills, more prevalent dependency on subsistence agriculture, physical and social isolation, specific disadvantages which become linked to ethnic identities and a greater exposure to natural disasters and risks [@Kozel2014].

The country's overall estimated Gini coefficient for income fluctuates between 0.42 and 0.44 between 2010 and 2018, with the highest levels of income inequality observed in the Central Highlands in 2016,
though absolute income may be rising, with the top quintile having 9.2 times the income of the lowest quintile in 2010 and 9.8 times in 2016 [@Le2021].
On the other hand, the bottom 40% experienced a slight absolute rise in mean income per capita from 4.00 USD (2011 PPP) in 2014 to 5.00 USD (2011 PPP) in 2018 [@WorldBank2022e].
For Gini coefficients estimated using consumption per capita, see @fig-vnm, which shows similar trends of increasing inequality, with 2010 constituting a significant increase.
Economic inequality and poverty in Vietnam thus underlies an intersectional focus, between ethnic minorities, regional disparities, rural-urban divides and gendered lines,
one which exogenous shocks can rapidly exacerbate as the example of the COVID-19 pandemic has recently shown [@Ebrahim2021].

```{python}
#| label: fig-vnm
#| fig-cap: "Gini index of consumption per capita for Vietnam"
gni_cnsmpt = vnm[vnm['resource'].str.contains("Consumption")]
gni_cnsmpt = gni_cnsmpt[gni_cnsmpt['scale'].str.contains("Per capita")]
gni_cnsmpt = gni_cnsmpt[gni_cnsmpt['source'].str.contains("World Bank")]
gni_cnsmpt = gni_cnsmpt[gni_cnsmpt['areacovr'].str.contains("All")]
gini_plot(gni_cnsmpt)
```

::: {custom-style="caption"}
Source: Author's elaboration based on UNU-WIDER WIID (2022).
:::

<!--
* estimated Gini coeff, overall income distribution: [@Le2021]
    * fluctuating 0.42-0.44 (2010-2018)
    * highest in Central highlands (2016)
    * absolute income inequality may be rising, top quintile 9.2 times income lowest quintile (2010) to 9.8 times (2016) (Hung 2019)
-->

<!-- rural inequality -->
In the 1990s, as the initial stages of the Doi Moi reform bore fruit with economic growth,
the first amplifications of inequalities along new rural-urban boundaries became visible.
There are two complementary views on the primary dimensions of rural inequalities.
On the one hand, the urban-rural divide may be driven by structural effects:
the welfare returns to education and agricultural activities changed dramatically from,
and with it the requirements on policy adaptations required for stemming inequality.

Nguyen et al. [-@Nguyen2007] argue this for the period of 1993-1998, with their findings that income returns to education improved dramatically over this time and arguing through this that suggested development policies had a strictly urban bias ---
on the whole they would benefit both from better education and vastly benefit from the restructuring of Vietnam's economy.
This view was in turn confirmed when Theil Index decomposition found within-sector inequality remaining largely stable while between-sector inequality rose significantly [@Fesselmeyer2010].
On the other, Thu Le and Booth [-@ThuLe2014] argue that the urban-rural inequality continued to increase over the years due to both covariate effects and the returns to those covariate effects, primarily education age structures and labor market activities, but also geographic location.

The gap between urban and rural sectors grew, a gap which would continue to widen until 2002, when within-sector rural inequalities started to become more important for inequalities than those between the sectors [@Fritzen2005; @ThuLe2014].
In the time of within-sector inequality becoming more pronounced many studies, while important contributions to continued inequality research, had a tendency to mask those inequalities in favor of continued analysis of between-sector trends ---
often to the detriment of the high degree of heterogeneity depending on geographic characteristics such as remoteness or cultural factors, as Cao and Akita [-@Cao2008] note.

In a recent study, Bui and Imai [-@Bui2019] build on this earlier work,
and also find access to basic education the linchpin of improving rural welfare while its lack combined with economic restructuring precluded many from equal opportunities toward human capital improvement.
They find that, as within-sector became more pronounced again after 2010,
the large proportion of uneducated heads of households in rural sectors and low social mobility of rural poor combine to increase within-sector inequality while the economy overall changing toward salaried work compounded within-rural and urban-rural disparities.
Early income studies generally highlighted the important role of agricultural incomes in reducing, or at the very not exacerbating, income inequality [@Benjamin2004].
Benjamin et al. [-@Benjamin2017] expand on this over a longer time-frame by decomposing different household income sources underlying Vietnam's structural economic changes.
They find that, while there is an overall decrease in income inequality throughout Vietnam between 2002 and 2014 and the urban-rural divide also continued its downward trend,
rural inequality indeed increased over this time.

Wage income and family business income were the main drivers of overall inequality in 2002 (accounting for over 30% of income but 60% of inequality) and remittances add a small share on top,
which, while decreased in effect (risen to 42% of total income),
remain majorly correlated with income distributions and thus income inequality.
@Bui2019 confirm this with a per capita income Gini coefficient of 0.36 to 0.39 between 2008 and 2010 which,
decomposed into Theil indices for between rural and urban and within rural sectors show that rural-urban inequalities are smaller and decreasing,
while within-rural inequalities are large and increasing.
Thus, while the study points to both more prevalent and equally distributed labor markets and wage labor opportunities,
these effects apply to the overall population and not just within-rural inequalities which are driven in large part by ethnicity, education and environmental factors.

<!--
* estimated Gini coeff, rural and urban expenditure: [@Bui2019]
    * 0.36-0.39 (2008-2010), 0.36 (2012)
    * Theil rural-urban between: 0.22-0.19 (2008-2012)
    * Theil rural-rural within: 0.74-0.83 (2008-2012)
-->

<!-- ethnicity inequality -->
Ethnic minorities in Vietnam are distinctly over-represented in poverty in addition to often being left behind in the development process, not least due to being extreme representatives of the economic situation of Vietnam's rural population.
Ethnic minority households have a tenuous economic position - and it is deteriorating.
In earlier studies on ethnic inequalities in Vietnam, a strong welfare gap between ethnic minorities and the majority was already visible.
Van de Walle [-@vandeWalle2001] reports the situation of ethnic minorities inhabiting predominantly remote rural areas with lower living standards than the ethnic majority,
a finding he suggests waws created due to environmental and structural differences (difficult terrain, poor infrastructure, less access to off-farm work and the market economy and inferior access to education) and compounded by social immobility and social isolation.
Baulch et al. [-@Baulch2012] find that between 1993 and 2004, the welfare gap between ethnic minorities and the ethnic majority had increased by 14.6%, two-fifths of which were due to endowments such as demographic structure and education while geographic variables make up less than one-fifth.
They additionally suggest some drivers of the inequality being the lack of ability speaking the Vietnamese language or the distance to a commune or district center amplifying isolating effects, though a large part of the change was linked to temporal changes of unobservable factors -
which the study conjectures to be due to negative ethnic stereotyping, a poor understanding of ethnic customs and culture and further (unobserved) variations in household-level endowments.

While in 2002 the ethnic minority population living in rural areas was below 15%, it rose to over 18% in 2014 - both due to higher fertility among minorities and ethnic majority Kinh urbanizing at a higher rate - and the ratio of Kinh to minority incomes rose to more than 2.0 in 2014 [@Benjamin2017].
The same study finds that income inequality rose even more sharply *within* ethnic minorities, while that of rural Kinh, though increasing from 2002 to 2014, fell back to 2002 levels around 2014.
These findings suggest that the primary drivers of rural income inequality are a growing gap between Kinh and minorities while at the same time a similar rising inequality develops among minority rural populations themselves.
<!-- TODO Find levels of population rural/urban in other sources -->
In the same vein as the urban-rural divide, Nguyen et al. [-@Nguyen2007] thus argue for structural policy failures which essentially lowered the returns on ethnicity along sectorial dividing lines of education and primary income types.

<!--
* estimated Gini coeff: rural ethnicities [@Baulch2012]
    * per capita HH expenditure between Kinh and Hoa majority 0.27 (1993, 2004)
    * per capita HH expenditure within minority groups 0.24 (1993), 0.29 (2004) sg increase
-->

### Natural disasters and inequalities

<!-- environmental inequality -->
While the effect of agriculture on inequality outcomes is an equalizing one,
its future growth, and that of agricultural livelihoods, is threatened by vulnerability to risks such as natural disasters and environmental degradation, exacerbated through climate change [@Kozel2014].
Kozel [-@Kozel2014] goes on to argue the continuous precarity of poor households against economy-wide shocks (such as the effect of climate change on rainfall and temperatures) but also highlights the danger of vulnerable households *falling* into poverty through generated inequalities.

Looking at the particularities of flood risk management in the Ninh Binh province, Mottet and Roche [-@Mottet2009] find that most areas within the region are vulnerable.
They find the strengths of current management lying in prevention with existing dykes designed to channel high waters, effective monitoring of weather conditions (rainfall or typhoons) and consolidation or elevation of existing residences, while the weaknesses are mainly centered around insufficient information given to inhabitants over flood risks, few compensation systems for flood victims and construction policies continuing to allow building in flood-endangered zones.

Sen et al. [-@Sen2021] estimate that the main barriers to better information are farmers' lack of trust toward formal climate-related services, their lack of perceived risk from climate change itself and difficulties in balancing both climate adaptation and economic benefits of interventions.
They argue that, while ethnicity itself is not a barrier to information access with all farmers receiving information through informal channels ---
friends neighbors and market actors instead of agricultural departments or mass media ---
cultural issues such as language do come into play and act as a barrier.
Reactionary economic mitigation efforts by households, such as reduced healthcare spending, selling of land or livestock assets, taking children out of school due to needing assistance at home can in turn lead to longer-term adverse consequences (thus, *mal-adaptation*) [@Kozel2014].

<!--
* estimated Gini coeff: Ninh Binh province [@Kozel2014]
    * 0.283 (2013)
    * top 90th percentile 3.57 times income 10th percentile
    * 13.63 percent in richest quintile
-->

<!-- extreme events / climate change -->
The results are further intensification of inequality along existing social lines during extreme events such as flooding:
The effects of inequalities mainly affecting ethnic minorities are illustrated by Son and Kingsbury [-@Son2020],
with droughts impacting yield losses between 50% and 100%, cold snaps leading to loss of livestock and floods damaging residential structures but even more importantly disrupting livelihoods through landslides, crop destruction and overflowing fish ponds.
Locally employed coping strategies, they argue, are always conditional on the strength and foresight of institutions and implemented preventative policies along local but also regional and central levels.

Similarly, Ylipaa et al. [-@Ylipaa2019] analyze impacts mainly across the gender dimension to find that,
resulting inequalities may be exacerbated with differentiated rights and responsibilities leading to unequal opportunities and, especially, decreased female mobility in turn increasing their vulnerability to climate impacts with a reduced capacity to adapt.
Hudson et al. [-@Hudson2021] along the same dimension find that,
while the set of relevant variables is largely similar with age, social capital, internal and external support after the flood and the perceived severity of previous flood impacts having major impacts,
women tend to show longer recovery times and psychological variables can influence recovery rates more than some adverse flood impacts.

While the quantitative evidence for impacts of such shock events are relatively sparse, Jafino et al. [-@Jafino2021] lament the overuse of aggregate perspectives,
instead disaggregating the local and inter-sectoral effects to find out that flood protection efforts in the Mekong Delta often predominantly support large-scale farming while small-scale farmers can be harmed through them.
They find that measures decrease the aggregate total output and equity indicators by disaggregating profitability indicators into inundation, sedimentation, soil fertility, nutrient dynamics and behavioral land-use in an assessment which sees within-sector policy responses often having an effect on adjacent sectors,
increasing the inter-district Gini coefficient.

Adaptation during these catastrophic events reinforces the asset and endowment drivers of non-shock event times,
with impacts levels often depending on access to non-farm income sources, access to further arable land, knowledge of adaptive farming practices and mitigation of possible health risks such as water contamination [@Son2020].
Karpouzoglou et al. [-@Karpouzoglou2019] make the point that, ultimately, the pure coupling of flood resilience into infrastructural or institutional interventions needs to take care not to amplify existing inequalities through unforeseen consequences ('ripple effects') which can't be escaped by vulnerable people due to their existing immobility.

<!-- conclusion -->
Inequality in Vietnam, then, is slowly rising across the whole population distribution,
runs the danger of increasing due to schisms opening within individual sectors of vulnerable groups.
Rural populations experience a trend towards increasing inequality within their sector,
driven primarily by the social exclusion and geographic isolation of ethnic minorities,
its most precarious population.
Ethnic minorities' inequality is slowly increasing due to receiving worse returns to their existing assets (especially human capital and land) and generally worse access to endowments in the first place (land and educational infrastructure).
The restructuring of the economy, turning the labor force toward urban areas and within them wage work in manufacturing and service industries,
leaves behind immobile rural populations whose ability to be employed for non-farm shrink further.

All these factors are at risk of experiencing large negative shocks as climate change exacerbates existing extreme environmental conditions,
which in turn threaten to increase economic inequalities for both the rural population at large, ethnic minorities and women especially.
Women in rural areas experience worse mobility and fewer economic opportunities and are thus less able to adapt to environmental degradation.
While inequality as an aggregate is kept relatively low Vietnam's growth rate,
both ethnic minorities and the rural female population are thus at risk of being left behind economically.

<!-- ## Script Issues
* Too wordy: on within-sector/between-sector rural dimensions; on non-quantitative environmental dimensions
* Too vague on 2 lowest quintile effects, especially as expressed through Gini impacts
* Wordings do not quite capture quintile poverty assessments for coming descriptive statistics
-->

<!-- development assistance -->
### Development assistance to Vietnam

```{python}
# Load CRS data
dfsub1 = pd.read_csv('data/raw/OECD_CRS/CRS1_Vietnam_11-13_05092022215007164.csv', parse_dates=True, low_memory=False)
dfsub2 = pd.read_csv('data/raw/OECD_CRS/CRS1_Vietnam_14-16_05092022215226180.csv', parse_dates=True, low_memory=False)
dfsub3 = pd.read_csv('data/raw/OECD_CRS/CRS1_Vietnam_17-20_05092022215427555.csv', parse_dates=True, low_memory=False)
df = pd.concat([dfsub1, dfsub2, dfsub3], ignore_index=True)
df = df.rename(columns={'\ufeff"DONOR"': 'DONOR'})
```

```{python}
#| label: fig-vnm-aid-financetype
#| fig-cap: "Total ODA for Vietnam per year, by financing type"
#| column: page
totals = df.loc[
    (df['SECTOR'] == 1000) & # Total
    (df['CHANNEL'] == 100) &
    (df['AMOUNTTYPE'] == 'D') &
    (df['FLOWTYPE'] == 112) &
    (df['AIDTYPE'] == "100") # contains mixed int and string representations
    ]
financetotals = totals.copy()
financetotals = financetotals[financetotals['DONOR'] < 20000] # drop all 'total' aggregations

## count amount of development aid financing instruments (grants/loans) by year and display
## count USD amount of development aid financing instumrnets by year and display
financetotals_grouped = financetotals.groupby(['Flow', 'Year']).agg({'Value': ['sum']})
financetotals_grouped = financetotals_grouped.reset_index(['Flow', 'Year'])
financetotals_grouped.columns = financetotals_grouped.columns.to_flat_index()
financetotals_grouped.columns = ['Financetype', 'Year', 'Value']

fig = px.line(financetotals_grouped, x='Year', y='Value', color='Financetype', labels={"Value": "Development aid amount, in millions of USD"}, markers=True, template="seaborn")
# 0-figure
fig.update_yaxes(rangemode="tozero")
fig.show()
```

::: {custom-style="caption"}
Note: Values shown are for all Official Development Assistance flows valid under the OECD CRS data, split into the type of financing flow, calculated as constant currency (2020 corrected) USD millions.
Source: Author's elaboration based on OECD ODA CRS (2022).
:::

Official Development Assistance (ODA) to Vietnam reached its highest point in 2014 with almost 5bn USD but generally decreased in the intervening years, as can be seen in @fig-vnm-aid-financetype.
Decreasing continuously after 2014,
development assistance reached its lowest point of the last ten years in 2019 when it fell to just under 2.5bn USD,
before increasing slightly to just above 2.5bn USD in 2020.
Development aid to Vietnam is primarily driven by ODA loans instead of ODA grants.

While grants were just under 1bn USD in 2011 and decreased slightly over the following years to 600m USD in 2019,
decreasing loans were also the primary driver of the overall development aid contributions,
with the overall monetary curve closely following that of loan contributions.
Thus, while loans constituted almost triple the USD amount of grants to Vietnam in 2011,
this number even climbed to almost 5 times the amount in 2014,
before falling to just over 2.5 times the amount of USD in loans compared to grants in 2020.
A large share of development aid contributions to Vietnam are also made up from other official flows[^officialflows],
a share which started equal to the absolute grant amount of USD contributed, rose steeply in 2013 to double the amount and fell equally steeply back to its original level in 2019.

[^officialflows]: Other official flows, per OECD CRS definition, describe contributions that do not meet the ODA criteria. These can include grants for primarily representational or commercial purposes, contributions having a grant element under the required share of 25% or primarily export-facilitating contributions. See https://doi.org/10.1787/6afef3df-en for a full definition.

```{python}
#| label: fig-vnm-aid-donortype
#| fig-cap: "Total ODA for Vietnam per year, separated by donor type"
#| column: page
totals = df.loc[
    (df['SECTOR'] == 1000) & # Total
    (df['FLOW'] == 100) &
    (df['CHANNEL'] == 100) &
    (df['AMOUNTTYPE'] == 'D') &
    (df['FLOWTYPE'] == 112) &
    (df['AIDTYPE'] == "100") # contains mixed int and string representations
    ]
donortotals = totals.copy()
donortotals["Donortype"] = donortotals["DONOR"].map(donortypes)
donortotals = donortotals[(donortotals["Donortype"] != "nondac")]

donortotals_grouped = donortotals.groupby(['Donortype', 'Year']).agg({'Value': ['sum']})
donortotals_grouped = donortotals_grouped.reset_index(['Donortype', 'Year'])
donortotals_grouped.columns = donortotals_grouped.columns.to_flat_index()
donortotals_grouped.columns = ['Donortype', 'Year', 'Value']
fig = px.line(donortotals_grouped, x='Year', y='Value', color='Donortype', labels={"Value": "Development aid, in millions"}, markers=True, template="seaborn")
# 0-figure
fig.update_yaxes(rangemode="tozero")
fig.show()
```

::: {custom-style="caption"}
Note: Values shown are for all Official Development Assistance flows valid under the OECD ODA data, split into bilateral development donor countries (dac) and multilateral donors (mlt), as constant currency (2020 corrected) USD millions.
Source: Author's elaboration based on OECD ODA CRS (2022).
:::

Bilateral donor contributions make up the largest part of development aid contributions to Vietnam, as can be seen in @fig-vnm-aid-donortype.
Both bilateral and multilateral contributions increase from 2011 to 2014 and subsequently begin decreasing.
While bilateral contributions do not increase in absolute amounts afterwards until 2020,
multilateral contributions increase again from 2019 to 2020.
Nevertheless, bilateral contributions are consistently higher than multilateral,
having around a 1.5 times higher share of absolute USD contribution,
though growing to just over 2 times the share in 2017,
before quickly shrinking down to just 1.3 times the share of multilateral contributions in 2018.
This gap may close further in the future, with multilateral contributions being on an increase and bilateral contributions still decreasing.

```{python}
#| label: tbl-vnm-aid-water
#| tbl-cap: "ODA projects of water supply and risk reduction in Vietnam per year, separated by project type"
#| column: page

# 14020-22 - large scale potable water
# 14030-32 - individual-level water and sanitation supply
# 14081 - education and training in water supply
totals = df.loc[
    (
        (df['SECTOR'] == 14020) |
        (df['SECTOR'] == 14021) |
        (df['SECTOR'] == 14022) |
        (df['SECTOR'] == 14030) |
        (df['SECTOR'] == 14031) |
        (df['SECTOR'] == 14032) |
        (df['SECTOR'] == 43060)
    ) &
    (df['CHANNEL'] == 100) &
    (df['AMOUNTTYPE'] == 'D') &
    ((df['FLOW'] == 11) | (df['FLOW'] == 13)) & # Total
    (df['FLOWTYPE'] == 112) &
    (df['AIDTYPE'] == "100") # contains mixed int and string representations
    ]
floodaid = totals[totals['DONOR'] < 20000] # drop all 'total' aggregations

# Aggregate different measures of large scale water supply and sanitation
floodaid.loc[floodaid[(floodaid['SECTOR'] == 14021) | (floodaid['SECTOR'] == 14022)].index, 'SECTOR'] = 14020
floodaid.loc[floodaid[(floodaid['SECTOR'] == 14020)].index, 'Sector'] = "Large water supply and sanitation"
# Aggregate different measures of individual water supply and sanitation
floodaid.loc[floodaid[(floodaid['SECTOR'] == 14031) | (floodaid['SECTOR'] == 14032)].index, 'SECTOR'] = 14030
floodaid.loc[floodaid[(floodaid['SECTOR'] == 14030)].index, 'Sector'] = "Basic water supply and sanitation"

# Group by sector per year and sum all values
floodaid_grouped = floodaid.groupby(['Year', 'Sector']).agg({'Value': ['sum']})
floodaid_grouped = floodaid_grouped.reset_index(['Year', 'Sector'])
floodaid_grouped.columns = floodaid_grouped.columns.to_flat_index()
floodaid_grouped.columns = ['Year', 'Sector', 'Value']

crosstab = pd.crosstab(floodaid_grouped['Year'], floodaid_grouped['Sector'], margins=True, values=floodaid_grouped['Value'], aggfunc='sum')

# Rename and reorder columns
crosstab.columns = ['Basic water supply', 'Disaster risk reduction', 'Large water supply', 'All']
crosstab = crosstab[['Basic water supply', 'Large water supply', 'Disaster risk reduction', 'All']]

Markdown(tabulate(crosstab.fillna("0.00"), headers="keys", tablefmt="github", floatfmt=".2f"))
```

::: {custom-style="caption"}
Note: Values shown are for all Official Development Assistance flows valid under the OECD CRS data, calculated as constant currency (2020 corrected) USD millions. The categories under analysis are large- and small-scale water supply and sanitation infrastructure projects as well as disaster risk reduction which includes improved flooding prevention infrastructure.
Source: Author's elaboration based on OECD ODA CRS (2022).
:::

The breakdown of project-based development aid for water supply infrastructure and disaster risk reduction in Vietnam can be seen in @tbl-vnm-aid-water.
It shows the funds broken down into their use for three categories:
First, contributions to provide access to basic water supply and sanitation,
which subsumes building and maintaining handpumps, gravity wells, rainwater collection systems, storage tanks, and small, often shared, distribution systems.
Second, contributions to large-scale water supply and sanitation,
including potable water treatment plants, intake works, large pumping stations and storage, as well as the transmission and distribution through large-scale systems.
And last, contributions towards disaster risk reduction which is a larger umbrella concept aimed at building local and national capacities,
but includes infrastructure measures (e.g. flood protection systems),
preparedness measures (such as early warning systems),
and normative prevention measures (such as closer adherence to building and structural codes),
as well as risk transfer systems (insurance schemes or risk funds).
This constitutes the closest category to flood risk management itself,
which is part of the overarching disaster risk management dimension.

While overall aid contributions to Vietnam's water supply and risk management sectors have slightly increased over time from 206m USD in 2011 to their peak of 422m USD in 2016, they have largely stagnated around the level of 300m to 350m USD per year since then.
From the level of 96m USD in 2011,
access to basic water supply saw significant increases to its contributions from 2013 to 2016,
with 154m USD contributed at its peak in 2016 and shrinking drastically the following years to 39m USD in 2019,
its lowest contribution year.
Large water supply project contributions see a similar if less drastic curve, with contributions increasing from 105m USD in 2011 to 252m USD at their in 2018, before decreasing slightly over the next two years.

Thus, the contribution curves to basic and large-scale water supply projects somewhat follows the overall development aid contribution curve to Vietnam,
with peaks between 2016 and 2018 before more or less drastic drops in aid contributions.
Disaster risk reduction contributions, however, show the least similarity to the general trend,
with contributions being only 4m USD in 2011 before increasing year-over-year (with the exception of 2018) to reach their peak with 63m USD in 2020.
The most significant increases happened between the years 2014 and 2016, as well as again in 2020.
While the other contribution sectors follow a shrinking contribution in the years following 2014, then,
disaster risk reduction instead keeps on reaching an increase in its absolute contribution amounts,
perhaps pointing to a continued necessity for development in the sector.
